Author's response to reviews

Title: Posterior migration of Ahmed glaucoma valve tube in a patient with Reiger anomaly : A Case Report

Authors:

Vishnu S Gupta (me_vsgupta@gmail.com)
Harinder S Sethi (sethi2211@yahoo.com)
Malvika Gupta (dr.malvika@gmail.com)
Anuj Mehta (anujsun@hotmail.com)
Shivram Singh (drshiyram@yahoo.in)
pankaj Yadav (pankaj.diablo@gmail.com)
K.P S Malik (malikkps@rediff.com)

Version: 3 Date: 16 March 2010

Author's response to reviews:

To
Editor in Chief
BioMed Central

Respected Sir/Maam

As per your review, relevant major and minor revisions have been made to the manuscript including correction of author name list format and affiliations, addition of figure legends, correction in the format of references in the manuscript and minor typographical corrections as suggested.

Kindly review the same.

Warm Regards,

Dr.Malvika Gupta